Author's response to reviews

Title: Evaluating an in-home multicomponent cognitive behavioural programme to manage concerns about falls and associated activity avoidance in frail community-dwelling older people: Design of a randomised control trial [NCT01358032]

Authors:

Tanja AC Dorresteijn (t.dorresteijn@maastrichtuniversity.nl)
GA Rixt Zijlstra (r.zijlstra@maastrichtuniversity.nl)
Kim Delbaere (k.delbaere@powmri.edu.au)
Erik van Rossum (erik.vanrossum@maastrichtuniversity.nl)
Johan WS Vlaeyen (Johannes.Vlaeyen@psy.kuleuven.be)
Gertrudis IJM Kempen (g.kempen@maastrichtuniversity.nl)

Version: 2 Date: 18 July 2011

Author's response to reviews: see over
Dear Nina Titmus,

Thank you for assessing our manuscript.

I have resubmit the manuscript and the additional document regarding the funding.

We recently submitted a manuscript on the 6th of April 2011. Although this paper is not directly part of the RCT, we would like to mention it for your information:
Title: Older people’s preferences regarding formats of programs to manage concerns about falls.
Submitted.
Authors: Tanja A.C. Dorresteijn, G.A. Rixt Zijlstra, Yvonne J.J. Van Eijs, Johan W.S. Vlaeyen, Gertrudis I.J.M. Kempen

With kind regards,
Tanja Dorresteijn

---

**T.A.C. (Tanja) Dorresteijn, MSc**
PhD Candidate
Department of Health Services Research - Focusing on Chronic Care and Ageing -
t.dorresteijn@maastrichtuniversity.nl
www.maastrichtuniversity.nl/hsr

Mailing address: P.O. Box 616, 6200 MD Maastricht, The Netherlands Visiting address: Duboisdomein 30, 6229 GT Maastricht, Room 0.082 T +31 43 3882815 F +31 43 3884162

The Department of Health Services Research is part of the School for Public Health and Primary Care (CAPHRI) of the Faculty of Health, Medicine and Life Sciences.
www.caphri.nl

---

From: Nina Titmus [mailto:Nina.Titmus@biomedcentral.com] On Behalf Of BMC Editorial
Sent: donderdag 14 juli 2011 20:39
To: Dorresteijn T (ZW)
Subject: MS: 2080810683565871 Evaluating an in-home multicomponent cognitive behavioural programme to manage concerns about falls and associated activity avoidance in frail community-dwelling older people: Design of a randomised control trial [NCT01358032]

MS: 2080810683565871
Study protocol
Evaluating an in-home multicomponent cognitive behavioural programme to manage concerns about falls and associated activity avoidance in frail community-dwelling older people: Design of a randomised control trial [NCT01358032]
Tanja AC Dorresteijn, GA Rixt Zijlstra, Kim Delbaere, Erik van Rossum, Johan WS Vlaeyen and Gertrudis IJM Kempen
BMC Health Services Research

Dear Dr Dorresteijn,

Thank you for your submission to BMC Health Services Research. We have assessed your submission but before we can proceed please could we ask you to address the following:

- Please resubmit your manuscript without the additional files
- Please provide an additional document regarding the funding. Proof of external funding should include details of the amount of funding that was awarded for your study and should clearly state that the funding was approved.
- Could you also provide a list of any manuscripts that are under submission with other journals based on this study protocol

We look forwarded to hearing from you shortly.

With kind regards,
Nina

Nina Titmus
Scientific Editor
On behalf of Deesha Majithia
BioMed Central
236 Gray's Inn Road
London, WC1X 8HB

T: +44 (0)20 3192 2013
F: +44 (0)20 3192 2011
W: www.biomedcentral.com

Follow BioMed Central: